Author’s response to reviews

Title: The course of diabetes in children, adolescents and young adults: does the autoimmunity status matter?

Authors:

Rasa Verkauskiene (rasa.verkauskiene@kaunoklinikos.lt)
Evalda Danyte (evalda.danyte@gmail.com)
Rimante Dobrovolskiene (rimantedobra@gmail.com)
Ingrida Stankute (ingridutes@gmail.com)
Diana Simoniene (diana.kud@gmail.com)
Dovile Razanskaite-Virbickiene (doliukas@gmail.com)
Audrone Seibokaite (audroneshb@yahoo.com)
Brone Urbonaite (brone.u@takas.lt)
Nijole Jurgeviciene (nijole.jurgeviciene@gmail.com)
Astra Vitkauskiene (astra.vitkauskiene@kaunoklinikos.lt)
Valerie Schwitzgebel (Valerie.Schwitzgebel@unige.ch)
Dalia Marciulionyte (dalia.m@takas.lt)

Version: 2 Date: 04 Nov 2016

Author’s response to reviews:

1. Please note that all manuscripts must contain all the following sections under the heading 'Declarations':

   Abbreviations

   Ethics approval and consent to participate

   Consent for publication

   Availability of data and material

   Competing interests
Funding

Authors' contributions

Acknowledgements

Authors' information (optional)

Answer: corrected.

2. Please make sure the abstract follows the format outlined in our submission guidelines.

Answer: corrected.

3. Please change the "Introduction" heading to "Background".

Answer: corrected.

4. Please add a “Conclusions” section after the “Discussion” section. This should state clearly the main conclusions of the research article and give a clear explanation of their importance and relevance.

Answer: we have written "Conclusions" section and stated the main points of our research article.

5. Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them.

Answer: corrected.